Author's response to reviews

Title: Neonatal care in rural Karnataka: healthy and harmful practices, the potential for change

Authors:

Amy J Kesterton (kestertona@interactworldwide.org)
John Cleland (john.cleland@lshtm.ac.uk)

Version: 4 Date: 16 April 2009

Author's response to reviews:

Dear BioMed Central

I have made the formatting changes requested but there are two which I was not completely sure I understood correctly.

Tables - I have left the tables in the main article and removed the separate files from the submission system. Please let me know if this is not what was wanted.

Figure cropping - I am not exactly sure what is needed. I have removed the Figure title but currently left it the same size. Please let me know if you would like me to do anything else.

Many thanks and best wishes

Amy